Author's response to reviews

Title: Differences in relative risk of stillbirth among women of different regions of origin compared with women from Germany: an analysis of the German perinatal database, 2004-2007

Authors:

Anna Reeske (reeske@bips.uni-bremen.de)
Marcus Kutschmann (m.kutschmann@bqs-institut.de)
Oliver Razum (oliver.razum@uni-bielefeld.de)
Jacob Spallek (spallek@bips.uni-bremen.de)

Version: 4 Date: 22 August 2011

Author's response to reviews:

Dear editors,

we now included a statement which explains why ethical approval was not necessary in our study and we changed the format according to the journal style.

Best wishes,

Anna Reeske